Author's response to reviews

Title: Differences in muscle activity during hand dexterity tasks between women with arthritis and a healthy reference group

Authors:

Sofia Brorsson (<u>sbo@du.se</u>) Anna Nilsdotter (<u>Anna.Nilsdotter@regionhalland.se</u>) Carina Thorstensson (<u>carina.thorstensson@registercentrum.se</u>) Ann Bremander (<u>ann.bremander@spenshult.se</u>)

Version: 2 Date: 25 July 2013

Author's response to reviews: see over

DearEditor!

W e have m ade all the changes that you required. The changes are m arked in red in the m anuscript and a point by point are listed below.

1.) K indly rem ove the Title Page in the additional file and please incorporate it in the main manuscript file.DONE

2.) A lso, please add the "A bstract" section : DONE

3.) Please add "A cknow ledgem ents" Section: DONE

4.) Consent statem ent: Please state in the M ethods section w hetherw ritten inform ed consent for participation in the study w as obtained from participants or, w here participants are children, a parent or guardian. DONE

5.)W enote that Figure 1 contains individual patient in ages which could com prom ise anonym ity. Please clarify if you obtained consent from patients for publication of their in ages. If not this figure will need to be rem oved. We have obtained consent from patients for publication of these in ages.

All the best

Sofia Brorsson